%uncomment this line for English articles (don't forget to also uncomment \end{otherlanguage} at the end of this file!)
\begin{otherlanguage}{english}

\label{sec:example}

%title
\begin{center}
\bf
\Large
This is the title
\end{center}

%abstract
\begin{center}
\textit{
Use this style if an abstract is needed. 
}
\end{center}

%article
\section{Section} \label{name::sec::intro} %use this label for referencing the section
\subsection{Subsection}

References can be included like this: \Cref{name::sec::intro},  \cref{name::figure} and \Cref{name::table}. For footnotes use 
\footnote{\texttt{This is a footnote}}.
Figures should be included like \cref{name::figure} or \cref{name::figure2} and for tables use \cref{name::table}.
Citations can be included like this \cite{example::Einstein}.


%this is an example figure
%size can be changed by multiplying width with a constant factor, e.g. [width=\textwidth * 0.7]
\begin{figure}[h!]
\centering
\includegraphics[width=\textwidth]{example/figure.eps} %eps gets converted to pdf automatically
\caption{example figure}
\label{name::figure}

%example for combining multiple pictures
\begin{minipage}{0.45\textwidth}
\centering
\includegraphics[width=\textwidth]{example/figure.eps}
\end{minipage}
\begin{minipage}{0.45\textwidth}
\centering
\includegraphics[width=\textwidth]{example/figure.eps}
\end{minipage}
\caption{example figures}\label{name::figure2}
\end{figure}

%example table
\begin{table}[h!]
\centering
\begin{tabular}{|r|cc|}
\hline
A & B & C \\
\hline
Content A 1& Content B 1&  Content C 1 \\
Content A 2& Content B 2&  Content C 2 \\
\hline
\end{tabular}
\caption{Example table}
\label{name::table}
\end{table}

%name of author
\begin{flushright}
Your author
\end{flushright}


{
\small
\paragraph{Acknowledgements} 

If needed add acknowledgements here!
}

\begin{thebibliography}{99}
\bibitem{example::Einstein}
A. Einstein: \textit{Die Grundlage der allgemeinen Relativitätstheorie}. Annalen der Physik IV, 49:769-822, 1916.
\end{thebibliography}

\FloatBarrier
\clearpage

\end{otherlanguage}